

Global board reforms and corporate acquisition performance

LSE Research Online URL for this paper: http://eprints.lse.ac.uk/123947/

Version: Published Version

Article:

To, Thomas, Wu, Eliza and Zhao, Diya (2024) Global board reforms and corporate acquisition performance. Journal of Corporate Finance, 87. ISSN 0929-1199

https://doi.org/10.1016/j.jcorpfin.2024.102617

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

ELSEVIER

Contents lists available at ScienceDirect

Journal of Corporate Finance

journal homepage: www.elsevier.com/locate/jcorpfin



Global board reforms and corporate acquisition performance

Thomas To^a, Eliza Wu^{a,*}, Diya Zhao^{a,b}

ARTICLE INFO

Editor: M Bennedsen

JEL classification:

G30

G34

G38

K22

Keywords:
Board reforms
Mergers and acquisitions
Corporate governance
Investor protection

ABSTRACT

This paper examines the effect of board reforms on corporate acquisition performance using data from 31 countries. Using a difference-in-differences design, we find that the implementation of board reforms in the acquirer's country significantly increases acquirer returns. The increase is driven by reforms involving board independence, but not reforms involving audit committee independence nor the separation of CEO and board chair roles. Further analysis shows that the uplift in acquisition performance following improvements in board independence is strongest in acquirers with more agency problems. The 'Board reform strengthening' effect is concentrated in larger acquirers, with more free cash flows, executing large and public-target deals and operating in countries with ex-ante poor investor protection. The empirical evidence indicates that reforming board independence effectively alleviates agency problems between managers and shareholders and improves corporate acquisition performance.

1. Introduction

Agency problem

With serious corporate scandals and managerial misconduct that repeatedly surface around the world, many countries have realized the criticality of strengthening corporate governance systems. Corporate board reforms have been extensively implemented since the late 1990s and they include the requirements of board independence, audit committee independence, and the clear separation of board chair and CEO roles. In this paper, we examine the effect of these different types of board reforms around the world on corporate acquisition performance. As prior studies have established that entrenched managers often make self-serving acquisitions and this can result in the significant destruction of shareholder value (Masulis et al., 2007; Harford et al., 2012), it warrants this global investigation into whether the implementation of different types of board reforms can restrain managers from making self-serving acquisitions and thence improve corporate acquisition performance.

How board reforms affect corporate acquisition performance is a central empirical question that we address in this study as the relationship is not clear ex ante. On the one hand, agency theory suggests that raising corporate governance standards through strengthening board independence can alleviate agency problems between managers and shareholders and should thereby improve acquisition performance (hence we term this the 'Board reform strengthening' view). For example, Nguyen and Nielsen (2010) show that following the sudden death of independent directors, stock prices on average drop by 0.85%, which indicates that independent directors add value to firms. In the vein of alleviating agency conflicts, Guo and Masulis (2015) find that CEO turnover is more sensitive

E-mail addresses: thomas.to@sydney.edu.au (T. To), eliza.wu@sydney.edu.au (E. Wu), d.zhao12@lse.ac.uk (D. Zhao).

^a Discipline of Finance, University of Sydney Business School, The University of Sydney, Australia

^b Department of Finance, London School of Economics and Political Science, United Kingdom

^{*} Corresponding author at: Discipline of Finance, University of Sydney Business School, The University of Sydney, Codrington Street, Darlington, NSW 2006, Australia.

to firm performance in firms with independent boards, suggesting that independent directors are able to discipline entrenched managers. Lastly, most relevant to our study, Masulis and Zhang (2019) find that firms with more preoccupied (distracted) independent directors perform worse in corporate acquisitions, which suggests that independent directors are valuable in acquisition decisions.

However, on the other hand, critics of board reforms suggest that independent directors often face limited access to firm-specific information, which reduces their monitoring and advisory effectiveness (Raheja, 2005; Adams and Ferreira, 2007; Duchin et al., 2010; Masulis and Mobbs, 2011). Furthermore, others argue that independent directors lack sufficient industry knowledge and expertise needed for advisory and only hold small equity stakes in firms which limits their financial incentives (Harris and Raviv, 2008). Most relevant to our study, Faleye et al. (2011) document that firms with independent directors that monitor intensively exhibit worse acquisition performance, suggesting that the negative advisory effects associated with independent directors may outweigh the benefits of improved monitoring. These insights indicate that the implementation of board reforms may adversely affect acquisition performance (which we term the 'Board reform weakening' view). Therefore, whether global board reforms enhance or impede the success of corporate acquisitions remains an important empirical question.

We investigate this research question using a comprehensive sample of acquisitions from 31 countries around the world. We use the global board reforms identified by Fauver et al. (2017), which the authors identified from various sources including the World Bank, European Corporate Governance Institute, local stock exchange regulators, and prior academic studies. Acquisition performance is measured using cumulative abnormal stock returns around the acquisition announcement. The results from our baseline difference-indifferences (DID) regression analysis show that the implementation of global board reforms in the acquirer's country increases acquirer returns. The finding is robust to controlling for various time-varying country and firm-specific characteristics, as well as acquirer country fixed effects, target country fixed effects, industry fixed effects and year fixed effects. In terms of economic importance, we find that 5-day cumulative abnormal stock returns around the acquisition announcement are on average 1.94% higher after the implementation of board reforms, which translates into an increase of \$93 million in shareholder value for the average acquirer in our sample.

We further examine whether the documented effect of global board reforms on corporate acquisition performance is driven by potential nonparallel acquisition performance trends before these reforms. We conduct dynamic DID tests and find that the effect on acquisition performance is immediate and persists from the first year after the reform year but does not exist in the years prior to the reform. This empirical evidence lends support to a causal interpretation of the relationship between global board reforms and acquisition performance. To further sharpen our identification, we also conduct placebo tests by falsely assuming that the first board reform occurs in the two years before, one year before, one year after, or two years after the actual reform year. We do not find any significant effect on acquisition performance from these placebo tests. Taken together, the findings from our dynamic DID and placebo tests suggest that the parallel-trends assumption for the DID approach is satisfied and the documented effect of global board reforms on corporate acquisition performance is most likely causal.

We further conduct multiple robustness checks for the documented positive effect of global board reforms on corporate acquisition performance. First, we confirm that the positive effect of board reforms on acquisition performance is driven by the implementation of board reforms in the acquirer's country rather than the target's country. Second, we control for industry-year fixed effects in our results and find stronger effects suggesting that our results are not driven by industry economic shocks or merger waves. Third, we follow the suggestion by Baker et al. (2021) and end our sample before the final board reform in 2006 and find qualitatively similar effects. Fourth, we measure acquisition performance using narrower three-day cumulative abnormal returns rather than five-day cumulative abnormal returns and we find qualitatively similar results. Fifth, we also examine the effect of board reforms on profitability and sales growth for the acquirer, as value creation should be eventually realized if the acquisition truly creates value for shareholders. We indeed find a significant positive effect of board reforms on acquirers' profitability and sales growth.

Next, we examine which specific components of board reforms contribute to the increase in acquisition performance. We find that reforms on board independence, but not the requirement of audit committee independence nor the separation of the CEO and Chairman positions, drive the documented increase in acquisition performance. This result is consistent with the finding from Masulis and Zhang (2019) that firms with more preoccupied independent directors perform worse in corporate acquisitions.

Lastly, we examine the channels through which global board reforms increase acquisition performance. We reason that raising board independence can alleviate the well-documented agency problems between managers and shareholders and should thereby improve acquisition performance. To validate our reasoning, we conduct three additional tests.

First, we examine the types of acquirers that are driving the positive effect of board reforms on acquisition performance. Prior literature suggests that large firm size serves as an effective takeover defense as it is more difficult to acquire a large target (Masulis et al., 2007). In addition, managers in firms with large amounts of free cash flows are more likely to make self-serving acquisitions if not carefully monitored. We therefore expect that the positive effect of board reforms on acquisition performance should be driven by these firms as managers in these firms are more likely to conduct self-serving acquisitions without adequate monitoring by independent directors. Our results show that the positive effect of board reforms on acquisition performance is indeed stronger in larger firms and firms with higher levels of free cash flows.

Second, we examine the types of deals that are driving the positive effect of board reforms on acquisition performance. Prior studies suggest that entrenched managers are more likely to conduct self-serving large acquisitions due to empire-building motives (Masulis et al., 2007). Moreover, entrenched managers often avoid private targets as they are more likely to face large blockholder(s) in private firms that can monitor and discipline them going forwards (Harford et al., 2012). We find that the positive effect of board reforms on acquisition performance is driven by large deals and public-target deals, which is consistent with prior studies that suggest these deals are likely to be self-serving acquisitions conducted by managers who are not being carefully monitored.

Third, we examine the effect of board reforms on acquisition performance conditional on the level of investor protection in a country. Prior studies find that managers in countries with poor investor protection are more likely to extract private benefits because they are less likely to lose their jobs through the market for corporate control (Pinkowitz et al., 2006; Doidge et al., 2007). We therefore expect that the implementation of board reforms should be highly effective in these countries offering weak investor protection. Using shareholder rights and creditor rights as proxies for investor protection, we find that indeed the positive effect of board reforms on acquisition performance is concentrated in countries with poor investor protection.

Our study contributes to the relatively scant literature on the effect of independent director representation and corporate acquisition performance. Early research by Byrd and Hickman (1992) find using a sample of tender offers that independent boards increase bidder returns, but it fails to account for the endogeneity of board structure (Hermalin and Weisbash, 2003). More recently, papers that use shock-based settings to examine the causal effect of independent director representation and corporate acquisition performance provide mixed results. For example, Masulis and Zhang (2019) show that independent directors add value to acquisition decisions, while Faleye et al. (2011) document that independent directors are potentially detrimental to acquisition performance. Different from the prior studies which focuses only on U.S. acquirers, we use global board reforms as an exogenous shock to board independence to study the effect of independent director representation on corporate acquisition performance around the world. The role of legal institutions as external governance mechanisms has been shown to have an impact on the efficacy of internal governance structures (Leuz et al., 2003; Djankov et al., 2008), hence it is important to examine the effect of board independence in different countries. In this paper, we contribute to the literature by showing that raising independent director representation increases acquisition performance globally and the effect is unambiguously concentrated in countries with weak external governance mechanisms.

Our study also adds to the recent studies that investigate the effect of global board reforms on corporate outcomes. Fauver et al. (2017) document a positive effect of global board reforms on firm valuation. Chen et al. (2020) show that board reforms lead to a reduction in cash holdings. Hu et al. (2020) show that stock price crash risk reduces following board reforms. Li et al. (2020) show that the cost of equity increases for firms after the implementation of board reforms. Bae et al. (2021) find that firms pay higher dividends following board reforms. Ben-Nasr et al. (2021) document that board reforms reduce firms' bank debt ratios. Chen et al. (2022) show that board reforms are associated with a significant reduction in IPO underpricing. Hui et al. (2023) document an increase in loan spreads following board reforms. Surprisingly, there is a dearth of attention on examining the effect of global board reforms on mergers and acquisitions, the largest form of corporate investment. Board reforms are implemented with the aim of reducing the conflict of interests between managers and shareholders, and these potential conflicts are often the highest in mergers and acquisitions decisions because managers can receive large increases in compensation (which is often linked to firm size) and status if they can make self-serving empire building acquisitions. Therefore, it is surprising that no prior study has examined the effect of global board reforms on mergers and acquisitions, since the reduction in large value-destroying acquisitions is likely to be one of the key drivers underpinning the increase in firm value documented in Fauver et al. (2017). In this paper, we fill this void in the extant literature by documenting that the implementation of board reforms improves the performance of corporate acquisitions.

Our paper is most closely related to Kim and Lu (2013) showing that corporate governance reforms affect the ability of foreign acquirers to cherry-pick their targets in emerging markets. However, this study differs from theirs in that we detail the impact of board reforms on local acquirers' corporate acquisition performance and we show that not all board reforms are the same and it is specifically those reforms enhancing board independence that strongly support acquirers and improves their corporate acquisition performance. Furthermore, we highlight that the effectiveness of reforms that strengthen board independence are conditional on the efficacy of the institutional environment in protecting investors and that both the monitoring and advisory functions of independent directors are important. There are clear policy implications from our findings on the types of board reforms that policy makers and standard setters should collectively prioritize to improve corporate governance standards.

The remainder of the paper is organized as follows. Section 2 describes the data sources and research design. Section 3 presents the empirical results on the effect of global board reforms on corporate acquisition performance. Section 4 examines the channels through which global board reforms affect corporate acquisition performance. Section 5 concludes.

2. Data sources, sample selection and research design

In this section, we describe our data sources, sample selection, summary statistics, and research design.

2.1. Data sources and sample selection

The data used in this study are obtained from multiple sources. We begin by obtaining board reforms data compiled by Fauver et al. (2017), who identified board reforms data from 31 countries using various sources including the World Bank, European Corporate Governance Institute, local stock exchange regulators, and prior academic studies (Kim and Lu, 2013). For each board reform, Fauver et al. (2017) identify the reform year and reform component, which includes the requirement of board independence, audit committee independence and separation of the CEO and chairman positions. For countries with multiple board reforms, we focus on the first board reform since it represents a more plausibly exogenous shift towards better corporate governance. To reduce the concerns of confounding events, we follow Fauver et al. (2017) and restrict the sample to five years before and five years after each reform adoption (i.e., we only include acquisitions from five years before to five years after the acquirer country's reform implementation year). We also exclude acquisitions conducted during the reform year since the actual timing of the reform within a year is unknown to us.

Next, we collect corporate acquisitions data from the Securities Data Company (SDC) M&A database, country-level data from the

World Bank and firm-level data from Compustat and Compustat Global. We only include mergers and acquisitions driven by change in control, where acquirers control less than 5% of the target before the announcement and more than 50% after the transaction. We exclude cross-listed firms in our sample, as they often have to comply with the board governance related listing rules of multiple stock exchanges, which adds noise to our analysis. We also exclude firms located in the United Kingdom as the first board reform in the United Kingdom is implemented in 1992 and Compustat Global does not provide consistent coverage for UK firms in the five years before the reform. Our final sample consists of 5322 acquisitions conducted by 3118 unique acquirers from 31 countries over the time period from 1992 to 2011. Appendix 2 provides the board reform data for these 31 countries reproduced from Fauver et al. (2017), and Table 1 reports the sample distribution by country. The sample is geographically diverse, and the distribution of acquirers per country varies widely. For example, our sample consists of 1545 acquirers from the United States, and only 1 from Colombia.

We control for acquirer and target country-level characteristics as well as acquirer firm-level and deal-level characteristics when examining the effect of global board reforms on corporate acquisition performance. Our country-level controls include the acquirer country's and target country's level and growth rate of GDP, as the level of economic development influences acquisition decisions and target selection (Rossi and Volpin, 2004). Our acquirer firm-level controls include firm size, return on assets, book-to-market ratio and leverage ratio. Managers in large firms are naturally more entrenched and are more likely to make value-destroying acquisitions as firm size serves as an effective takeover defense for potential acquirers. Return on assets controls for the firm's profitability and performance. The effect of book-to-market is ambiguous as Lang et al. (1991) document a negative effect for tender offers and publicfirm acquisitions while Moeller et al. (2004) find a positive relation using their sample. The effect of leverage is also ambiguous as leverage on the one hand limits managerial discretion and provides incentives for managers to improve firm performance, while on the other hand managers in firms with higher leverage have incentives to play it safe and are more likely to undertake diversifying acquisitions that have negative announcement returns (Gormley and Matsa, 2016). Our deal level controls include relative deal size, a cross-industry deal indicator, a cross-border deal indicator, the method of payment, target public status and tender offer indicators. We winsorize all continuous variables at the 1st and 99th percentiles. Table 2 reports summary statistics. An average acquirer in our sample has a leverage ratio of 19%, book-to-market ratio of 0.66, and ROA of 7%. The average 5-day cumulative abnormal announcement return for our sample of corporate acquisitions is 1.25% and the size of the average acquisition is 29% of acquirer's market capitalization. Among the acquisitions, 43% are cross-industry deals, 14% are cross-border deals, 7% are tender-offer deals, and 50% are funded entirely by cash. Table 2 also shows that there are significant differences in acquirer firm characteristics, target firm characteristics, and deal characteristics between acquisitions conducted during the pre-reform period and the post-reform period. This is expected as acquirers and targets from different industries in different countries participate in the M&A market each year. These differences all drive differences in acquirer returns. Hence, we need to control for these firm characteristics, country characteristics, deal characteristics, along with country, industry, and year fixed effects in our multivariate analysis to gain a clear understanding on the effect of board reforms on corporate acquisition performance.

2.2. Research design

We exploit the staggered adoption of board reforms across countries and time to examine the effect of board reforms on corporate acquisition performance using a difference-in-differences (DID) design. Specifically, we estimate the following difference-in-differences (DID) regression model:

$$CAR \ (-2, +2)_{i,t} = \alpha + \beta Reform_{i,t} + \gamma Z_{i,j,k,t} + Acquirer Country_i + Target Country_k + Industry_m + Year_t + \varepsilon_{i,j,k,m,t}$$
 (1)

where i denotes firm, j denotes acquirer country, k denotes target country, m denotes industry and t denotes year. The dependent variable, $CAR (-2, +2)_{i,t}$, is the 5-day cumulative acquirer abnormal returns around the acquisition announcement date for firm i in year t. Abnormal returns are calculated by subtracting the acquirer country's stock market return from the acquirer's stock return. The board reform measure, $Reform_{j,t}$, is an indicator variable that takes the value of one for acquisitions conducted within the five years after a country's board reform, and zero for acquisitions conducted within the five years before a country's board reform. $Z_{i,j,k,t}$ is a vector of acquirer-country, target-country, acquirer firm and deal characteristics that are likely to affect a firm's acquisition performance. $AcquirerCountry_j$, $TargetCountry_k$, $Industry_m$ and $Year_t$ capture acquirer country, target country, industry and year fixed effects, respectively. We cluster standard errors by country as board reforms occur at the country-level.

The coefficient of interest is β . It captures the difference between changes in corporate acquisition performance in countries with

We only include deals in which acquirers control less than 5% of the target before the announcement to ensure that the acquirers were not blockholders in the target company before the acquisition, as blockholders often have the ability to influence managerial decisions. As a robustness check, we also re-run our analysis requiring the acquirer to control less than 10%, 20%, or 50% before the acquisition, and present the results in Appendix 3.

² We obtain cross-listing data from websites of major depositaries of ADRs; Bank of New York, Citibank, and J.P. Morgan.

³ As a robustness check, we also calculate cumulative abnormal returns (CARs) of the sample acquirers over the event window (-2, +2) around the announcement date in which the CARs are measured as returns in excess of those predicted by the market model with the benchmark being the country's index and parameters estimated over a period from 300 to 91 days prior to the announcement (Golubov et al., 2015). We present the regression results in Appendix 4.

As a robustness check, we also cluster standard errors by acquirer, industry and year. We present the regression results in Appendix 5.

Table 1Sample Distribution.

Country	Number of acquirers	Number of acquisitions
Australia	190	250
Belgium	3	3
Brazil	3	3
Canada	3	5
Chile	4	6
China	94	114
Colombia	1	2
Denmark	16	23
Finland	29	63
France	47	59
Germany	46	52
Greece	3	3
Hong Kong (China)	11	15
India	9	9
Indonesia	2	2
Italy	63	93
Japan	563	947
Mexico	2	3
Malaysia	170	228
Netherlands	14	18
Norway	37	69
Philippines	8	10
Poland	13	26
Portugal	2	3
Singapore	85	102
South Korea	43	55
Spain	19	23
Sweden	51	81
Switzerland	10	11
Thailand	32	44
US	1545	3000
Total	3118	5322

This table provides the number of acquirers and number of acquisitions by acquirer country.

Table 2
Summary Statistics.

	Mean	Std. dev.	Q1	Median	Q3	Pre-Reform	Post-Reform	Diff in Mean	Diff in Median
Reform	0.61	0.49	0.00	1.00	1.00				
CAR $(-2, +2)$ (%)	1.25	8.70	-3.15	0.44	4.91	0.96	1.43	-0.47	-0.33
Ln(GDP)	29.08	1.49	28.55	29.90	30.13	29.30	28.94	0.36***	0.77***
GDP Growth (%)	3.35	2.10	1.88	3.51	4.48	3.57	3.21	0.36***	1.62***
Target Ln(GDP)	28.89	1.51	28.13	29.84	29.99	29.08	28.78	0.30***	0.64***
Target GDP Growth (%)	3.42	2.23	1.88	3.51	4.48	3.56	3.33	0.23***	1.62***
Firm Size	5.99	2.15	4.50	5.83	7.30	6.37	5.76	0.61***	0.69***
Book-to-Market	0.66	1.21	0.22	0.42	0.70	0.63	0.67	-0.04	-0.07***
ROA	0.07	0.16	0.03	0.09	0.15	0.06	0.08	-0.02***	0.00
Leverage	0.19	0.18	0.03	0.15	0.30	0.18	0.20	-0.02***	-0.03***
Relative Deal Size	0.29	0.78	0.01	0.05	0.17	0.34	0.25	0.09***	0.04***
Cross-industry	0.43	0.50	0.00	0.00	1.00	0.39	0.46	-0.07***	0.00***
Cross-border	0.14	0.35	0.00	0.00	0.00	0.14	0.15	-0.01	0.00
Cash deal	0.30	0.46	0.00	0.00	1.00	0.20	0.37	-0.17***	0.00***
Private Target	0.50	0.50	0.00	1.00	1.00	0.47	0.52	-0.05***	-1.00***
Tender Offer	0.07	0.25	0.00	0.00	0.00	0.10	0.05	0.05***	0.00***
Takeover Premium (%)	36.48	35.72	14.63	30.68	51.05	42.37	30.37	12***	23.08***

This table provides summary statistics for the variables used in our baseline regressions. All variable definitions are provided in Appendix 1. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

board reforms and those without board reforms during the corresponding years. This approach classifies all firms in countries where board reforms are implemented into the treatment group, and those in countries where reforms are not yet implemented into the control group, which is commonly used in the literature (e.g., Bertrand and Mullainathan, 2003; Gormley and Matsa, 2016). If board reforms increase (decrease) corporate acquisition performance, consistent with the 'Board reform strengthening' ('Board reform weakening') view, we expect β to be significantly positive (negative).

3. Global board reforms and corporate acquisition performance

In this section, we examine the effect of global board reforms on corporate acquisition performance. We first present the difference-in-differences test results and then present the dynamic difference-in-differences, placebo, and robustness test results to rule out that our findings are driven by general macroeconomic conditions unrelated to board reforms. Furthermore, we examine how each of the three components of board reforms affect corporate acquisition performance.

3.1. Difference-in-differences test results

Table 3 presents the difference-in-differences (DID) test results that examines the effect of global board reforms on corporate acquisition performance. We start with a model that regresses abnormal returns on an acquirer country board reform indicator, acquirer and target country's level and growth rate of GDP, along with acquirer firm-level controls. Column 1 of Table 3 reports the results. We find that corporate acquisition performance increases following the implementation of a board reform in the acquirer's country, and the effect is significant at the 10% level. Next, we add deal-level controls into the regression and report the results in Column 2 of Table 3. The positive effect of global board reforms on corporate acquisition performance becomes significant at the 5% level. Finally, we exclude countries with less than 10 unique acquirers in our sample and find that the effect of global board reforms on corporate acquisition performance remains positive and significant at the 5% level, as reported in Column 3 of Table 3. In terms of economic importance, we find that 5-day cumulative acquirer abnormal stock returns around the acquisition announcement are on average 1.94% higher after the implementation of board reforms, which translates into an increase of \$93 million in shareholder value

Table 3Global board reforms and corporate acquisition performance.

Dependent Variable	CAR $(-2, +2)$		
	Full Sample		Exclude countries with less than 10 unique acquirers
Reform	1.772*	1.941**	2.159**
	(1.985)	(2.163)	(2.384)
Ln(GDP)	2.979*	3.434*	3.235*
	(1.929)	(2.037)	(1.830)
GDP Growth	0.005	0.001	-0.008
	(0.021)	(0.006)	(-0.034)
Target Ln(GDP)	1.389	1.319	1.632
	(0.944)	(0.863)	(1.020)
Target GDP Growth	-0.142	-0.138	-0.152
	(-0.637)	(-0.595)	(-0.629)
Firm Size	-0.655***	-0.569***	-0.571***
	(-14.241)	(-8.959)	(-8.322)
Book-to-Market	-0.077	-0.023	0.004
	(-0.637)	(-0.179)	(0.030)
ROA	0.481	0.256	0.465
	(1.209)	(0.578)	(0.931)
Leverage	0.380	0.599	0.570
0	(0.794)	(0.934)	(0.884)
Relative Deal Size	, , ,	-0.631***	-0.653***
		(-6.852)	(-8.338)
Cross-industry		-0.194	-0.185
,		(-1.133)	(-1.111)
Cross-border		-0.743*	-0.688
		(-1.814)	(-1.586)
Cash deal		0.350	0.304
		(1.070)	(0.881)
Private Target		0.096	0.103
		(0.163)	(0.172)
Tender Offer		0.551	0.563
Tender offer		(1.352)	(1.382)
Acquirer country FE	Yes	Yes	Yes
Target country FE	Yes	Yes	Yes
Industry FE	No	Yes	Yes
Year FE	Yes	Yes	Yes
N	5322	5322	5272
Adjusted R ²	0.060	0.064	0.061

This table presents the effect of global board reforms on corporate acquisition performance. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

for the average acquirer in our sample. Hence, our results are more in line with the 'Board reform strengthening view' with regards to corporate acquisition performance.

The coefficients of our control variables exhibit the expected signs. For example, managers in larger firms are naturally more entrenched and therefore are more likely to make value-destroying acquisitions, large deals are more likely to be value-destroying as these deals are likely to be conducted due to the empire-building motives of the CEO, and stock prices of acquirers in cross-border deals often respond negatively to acquisition announcements as investors are concerned that acquirers usually have worse information than the sellers in cross-border deals (Moeller and Schlingemann, 2005).

Table 4Global board reforms and corporate acquisition performance: dynamic DID and Placebo Tests.

Panel A: Dynamic DID	Test				
Dependent Variable	CAR (-2, +2)				
	(1)	(2)	(3)	(4)	(5)
	Dynamic DID test	Placebo reform 2 years before actual reform	Placebo reform 1 year before actual reform	Placebo reform 1 year after actual reform	Placebo reform 2 years after actual reform
Year (−1)	0.199				
	(0.417)				
Year (+1)	2.096*				
* * *	(2.003)				
Year (+2)	2.207*				
	(1.755)				
Pseudo Reform	(=1, ==)	0.287	0.710	0.267	-0.177
rocado recom		(0.256)	(0.727)	(0.355)	(-0.230)
Ln(GDP)	3.504**	3.504**	0.708	4.202*	2.906**
	(2.118)	(2.118)	(0.195)	(1.897)	(2.127)
GDP Growth	-0.005	-0.005	0.084	0.090	0.232
dbi diowiii	(-0.019)	(-0.019)	(0.318)	(0.364)	(0.945)
Target Ln(GDP)	1.223	1.223	0.345	-1.456	0.469
Target Lin(GDF)	(0.827)	(0.827)	(0.162)	(-1.040)	(0.343)
Target GDP Growth	-0.138	-0.138	-0.148	-0.208	-0.318
Target GDP Growth					
Pi Ci	(-0.594)	(-0.594) -0.570***	(-0.968)	(-0.870)	(-1.452)
Firm Size	-0.570***		-0.534***	-0.608***	-0.590***
D 1 . 35 1 .	(-9.013)	(-9.013)	(-7.714)	(-6.225)	(-7.218)
Book-to-Market	-0.025	-0.025	0.043	0.045	0.090
	(-0.198)	(-0.198)	(0.319)	(0.269)	(0.891)
ROA	0.255	0.255	0.618	0.240	-0.350
	(0.574)	(0.574)	(1.393)	(0.611)	(-0.671)
Leverage	0.597	0.597	0.651	0.123	0.517
	(0.929)	(0.929)	(0.775)	(0.161)	(0.625)
Relative Deal Size	-0.630***	-0.630***	-0.723***	-0.687***	-0.620***
	(-6.851)	(-6.851)	(-8.882)	(-7.940)	(-4.494)
Cross-industry	-0.192	-0.192	-0.189	-0.055	-0.258
	(-1.115)	(-1.115)	(-1.091)	(-0.231)	(-1.282)
Cross-border	-0.738*	-0.738*	-0.699	-1.247***	-0.563
	(-1.789)	(-1.789)	(-1.352)	(-2.825)	(-1.352)
Cash deal	0.355	0.355	0.180	0.077	0.135
	(1.096)	(1.096)	(0.997)	(0.253)	(0.414)
Private Target	0.096	0.096	0.519	0.300	0.090
	(0.163)	(0.163)	(0.892)	(0.379)	(0.177)
Tender Offer	0.551	0.551	-0.247	0.182	0.159
	(1.348)	(1.348)	(-0.563)	(0.353)	(0.310)
Acquirer country FE	Yes	Yes	Yes	Yes	Yes
Target country FE	Yes	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes	Yes
N N	5322	4100	4861	5759	5644
Adjusted R ²	0.064	0.071	0.071	0.072	0.063

This table presents the dynamic DID regression results and placebo DID regression results on the effect of global board reforms on corporate acquisition performance. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. For k being -1, 0 or 1, Year(k) is an indicator variable that equals one if year t is the kth year relative to the adoption year of the first board reform of the acquirer country and equals zero otherwise. Year (+2) is an indicator variable that equals one if year t is the 2nd year onward after the adoption year of the first board reform of the acquirer country and equals zero otherwise. We use the first four pre-reform years (years -5 to -2) of the country as the reference years. Panel B reports the placebo DID regression results. Placebo Reform is an indicator variable that equals one for the years after a pseudo reform and equals zero for the years before the pseudo reform. Appendix 1 presents detailed definitions and constructions of the other variables. t-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

Table 5Global board reforms and corporate acquisition performance: robustness tests.

Dependent variable		CAR (-2, +2)			CAR (-1, +1)	ΔIndustry- Adjusted ROA	ΔIndustry- Adjusted Sale Growth	
(1)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
	Control for Difference between Acquirer Reform and Target Reform	Only include cross- border deals and control for target-country board reform	Control for	End our sample in year 2005				
			industry-year fixed effects					
Reform	1.829*	3.683*	2.960***	2.325*	1.468**	0.021**	0.072**	
	(1.791)	(1.765)	(4.012)	(2.028)	(2.525)	(3.117)	(3.739)	
Ln(GDP)	3.533**	5.197	2.968**	4.396*	2.881***	0.001	-0.004	
	(2.096)	(1.390)	(2.298)	(1.818)	(2.895)	(0.281)	(-0.651)	
GDP Growth	0.002	-0.060	-0.078	-0.217	-0.122	0.000	-0.005**	
	(0.007)	(-0.209)	(-0.378)	(-1.578)	(-0.740)	(1.973)	(-5.401)	
Farget Ln (GDP)	1.136	2.855	1.704	-0.533	-0.205	-0.006	0.012	
(021)	(0.761)	(0.700)	(0.793)	(-0.136)	(-0.122)	(-1.955)	(2.147)	
Target GDP	-0.136	-0.208	-0.155	-0.133	-0.022	-0.001***	0.000	
Growth	**=**	**=**	**= **					
-	(-0.599)	(-0.938)	(-0.793)	(-0.932)	(-0.167)	(-4.869)	(0.091)	
Firm Size	-0.569***	-0.269	-0.539***	-0.553***	-0.413***	0.001***	-0.001**	
	(-9.017)	(-1.434)	(-10.621)	(-15.244)	(-6.819)	(5.918)	(-4.461)	
Book-to-	-0.022	0.133	0.041	0.157	0.097	0.001***	0.001	
Market								
	(-0.167)	(0.634)	(0.416)	(1.137)	(1.417)	(4.320)	(1.513)	
ROA	0.266	-3.172	-0.209	0.002	-0.217	-0.033***	0.010**	
	(0.580)	(-1.072)	(-0.378)	(0.003)	(-0.457)	(-17.001)	(4.333)	
everage	0.610	2.911*	1.014*	0.937*	0.277	0.003***	0.004**	
	(0.974)	(1.813)	(1.955)	(1.833)	(0.238)	(12.710)	(4.402)	
Relative Deal Size	-0.630***	-1.865***	-0.659***	-0.882***	-0.753***	-0.000***	0.001**	
	(-6.794)	(-4.481)	(-10.228)	(-11.167)	(-9.405)	(-27.407)	(5.600)	
Cross- industry	-0.194	-0.919**	-0.319	-0.418*	-0.089	-0.000***	0.001	
ilidustry	(-1.138)	(-2.086)	(-1.633)	(-1.857)	(-0.671)	(-6.203)	(1.075)	
Cross-border	-0.767*	(2.000)	-0.842*	-0.603	0.026	-0.000	-0.004*	
oroso border	(-1.760)		(-1.887)	(-0.947)	(0.067)	(-0.440)	(-2.893)	
Cash deal	0.350	-0.089	0.363	0.113	0.245	0.002***	0.001**	
dioir dear	(1.067)	(-0.103)	(0.960)	(0.221)	(0.617)	(9.580)	(3.949)	
Private	0.099	-1.576***	0.216	0.572	0.076	-0.001**	-0.003*	
Target	0.000	110, 0	0.210	0.07.2	0.070	0.001	0.000	
ruiget	(0.170)	(-2.887)	(0.354)	(0.762)	(0.139)	(-3.710)	(-2.986)	
Tender Offer	0.551	-0.814	1.163**	1.537***	0.668**	-0.000	0.001	
	(1.359)	(-0.770)	(2.217)	(3.496)	(2.273)	(-0.728)	(2.245)	
Difference in Reform	0.569	(,		(,	, , , ,	,,	()	
10101111	(0.415)							
Γarget-	(0.113)	-1.389						
country Reform								
VEIOLIII		(-0.981)						
Acquirer	Yes	(-0.981) Yes	Yes	Yes	Yes	Yes	Yes	
country	165	165	165	165	165	165	163	
ге. Farget	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
country	10	103	100	103	103	103	10	
FE nductor FF	Voc	Voc	No	Voc	Voc	Voc	Voc	
ndustry FE	Yes	Yes	No No	Yes	Yes	Yes	Yes	
rear FE	Yes	Yes	No Vec	Yes	Yes	Yes	Yes	
ndustry-year FE	No	No	Yes	No	No	No	No	
N	5322	673	5165	3538	5324	3724	3562	
Adjusted R ²	0.060	0.182	0.141	0.142	0.064	0.223	0.140	

This table presents the robustness test results for the effect of global board reforms on corporate acquisition performance. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement, CAR (-1, +1) is the three-day cumulative abnormal return around the acquisition announcement, Δ Industry-Adjusted ROA is the change in acquirer's industry-adjusted ROA from year before the acquisition to the year after the acquisition. Δ Industry-Adjusted Sales Growth is the change in acquirer's industry-adjusted sales growth from the year

before the acquisition to the year after the acquisition. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

3.2. Dynamic DID and placebo tests

We further conduct dynamic DID and placebo tests to assess whether our results may be driven by concurrent changes unrelated to the board reforms. In our dynamic DID model, we replace the reform indicator with indicator variables that track the effect of board reforms before and after they are adopted and become effective. Specifically, we include the indicator variables Year (-1), Year (+1) and Year (+2). Year (-1) equals one for the year before the reform, and zero otherwise. Year (+1) equals one for the year after the reform, and zero otherwise. Year (+2) equals one for the second and subsequent years after the reform, and zero otherwise. Column 1 of Table 4 presents the results. The coefficient estimate of Year (-1) is insignificantly different from zero, which suggests that the implementation of board reforms are not in response to changes in economic conditions that affected corporate acquisition performance. Moreover, we find that the positive effect of board reforms on corporate acquisition performance materializes in the first year after the implementation of a board reform. Specifically, the results show that cumulative abnormal announcement returns on average increases by 2.1% in the first year after the implementation of a board reform, and the effect lasts from the second year to the fifth year, with the increase in cumulative abnormal announcement returns increasing to 2.2%.

We also conduct a series of placebo tests by falsely assuming that the board reform occurs in the 2 years before the actual reform year, 1 year after the actual reform year, or 2 years after the actual reform year. Similar to our main sample specification, we restrict our sample to between 5 years before and 5 years after each pseudo reform. Columns 2 to 5 of Table 4 report the placebo test results. We do not find any significant effect on acquisition performance from these pseudo reforms. Taken together, the findings from our dynamic DID tests and placebo tests suggests that the documented effect of global board reforms on corporate acquisition performance is most likely to be causal.

3.3. Additional robustness tests

We further conduct multiple robustness checks for the documented positive effect of global board reforms on corporate acquisition performance and present our findings in Table 5. First, there is a possibility that our results are driven by the target country's adoption of a board reform, as in 86% of the deals in our sample the target firm is located in the same country as the acquirer. An alternative interpretation of our results is that board reforms reduces the ability of targets to put up a takeover defense and therefore lowers the cost of acquisition and thereby increases acquirer returns. We examine the effect of board reforms on takeover premium and report the results in Appendix 6. We do not find a significant effect of board reforms on takeover premium. There is also a possibility that the difference in the level of investor protection between the acquirer and the target is driving our results. To address this concern, we control for the relative difference in board reforms between the two countries and report our results in Column 1 of Table 5. We continue to find a significant positive effect of global board reforms on corporate acquisition performance. To further examine whether our results are driven by the target country's adoption of a board reform, we restrict our sample to include only cross-border deals and control for target countries' implementation of board reforms in our regressions. Column 2 of Table 5 presents the results. We find that the documented positive effect of global board reforms on corporate acquisition performance is driven by the acquirer country's implementation of board reforms and not the target country's implementation of board reforms. Overall, these results make sense as managers often hold equity positions in their firms and hence managers in the target firm would act in the best interest of its shareholders when facing a takeover bid regardless of whether they are being carefully monitored or not.

Second, there is a potential concern that the documented positive effect of global board reforms on corporate acquisition performance may be driven by merger waves that occurred in the years around the implementation of board reforms, as Rhodes-Kropf and Viswanathan (2004) document that acquisitions are often overvalued during merger waves. Given that industry-specific economic shocks cause merger waves (Mitchell and Mulherin, 1996), we control for merger waves by controlling for industry-year fixed effects in our regressions. Column 3 of Table 5 shows the results. We find even stronger positive effects of board reforms on corporate acquisition performance with this model specification, ruling out the possibility that our results are specific to reforms occurring during merger waves.

Third, we follow the suggestion by Baker et al. (2021) and end our sample before the final board reform in 2006, as the authors suggest that all countries in the sample have implemented board reforms by 2006 and hence there are no effective control firms after 2005 for the DID tests. Column 4 of Table 5 shows that the documented positive effect of board reforms on corporate acquisition performance remains qualitatively similar with this sample restriction.

Fourth, we use three-day cumulative abnormal returns rather than five-day cumulative abnormal returns to measure acquisition performance and we then reestimate the DID regressions in a sensitivity check. Column 5 of Table 5 shows that the results are qualitatively unchanged.

Fifth, we also examine the effect of board reforms on (industry-adjusted) profitability and sales growth (from year-1 to year+1) for the acquirer. These performance metrics are important because the value creation should be eventually realized if the acquisition truly creates value for shareholders, as suggested by Suk and Wang (2021) and Ben-David et al. (2022). Columns 6 and 7 of Table 5 presents the results. We indeed find a significant positive effect of board reforms on profitability and sales growth for the acquirer.

Overall, results from the robustness tests confirm the documented positive effect of global board reforms on corporate acquisition

performance, thereby lending further support for our 'Board reform strengthening' view.

3.4. Components of board reforms and corporate acquisition performance

Next, we examine how each of the three key types of board reforms affect corporate acquisition performance. First, we expect board reforms involving the requirement of board independence to have a positive effect on acquisition performance, as Masulis and Zhang (2019) find that firms with more preoccupied independent directors perform worse in acquisitions which indicates the importance of independent directors. Second, ex-ante the effect of audit committee independence reform on acquisition performance is ambiguous. On the one hand the existing literature documents that audit committee independence is generally related to better corporate transparency and less earnings management rather than investment decision-making (e.g., Carcello and Neal, 2002; Deli and Gillian, 2000; Klein, 2002), and hence we should not expect reforms involving audit committee independence to have a significant effect on corporate acquisition performance. However, on the other hand, Faleye et al. (2011) document that firms with independent directors that monitor intensively exhibit worse acquisition performance, suggesting that audit committee independence reform may have a negative effect on acquisition performance. Third, existing studies document that the existence of CEO-Chairman duality can both harm and improve investment decisions. On the one hand, agency theory suggests that the separation of the CEO and board chair roles help to improve corporate boards' monitoring effectiveness (Fama and Jensen, 1983; Jensen, 1993). On the other hand, organization theory suggests that having CEO-Chairman duality establishes strong and unambiguous leadership which enables the CEO to have

Table 6
Components of board reforms and corporate acquisition performance.

Dependent Variable	CAR $(-2, +2)$				
	(1)	(2)	(3)	(4)	
Board Independence Reform	1.329**			1.627**	
-	(2.212)			(2.486)	
Audit Committee Reform		0.844		-0.638	
		(0.912)		(-0.640)	
Separation of CEO-Chair Reform			0.532	0.580	
•			(0.300)	(0.314)	
Ln(GDP)	2.712	3.470*	3.085*	2.257	
	(1.618)	(2.008)	(1.899)	(1.419)	
GDP Growth	0.057	0.019	0.030	0.069	
	(0.253)	(0.081)	(0.132)	(0.295)	
Target Ln(GDP)	0.716	1.282	1.396	0.665	
	(0.483)	(0.828)	(0.954)	(0.470)	
Target GDP Growth	-0.132	-0.137	-0.139	-0.131	
o .	(-0.559)	(-0.586)	(-0.596)	(-0.555)	
Firm Size	-0.571***	-0.567***	-0.567***	-0.572***	
	(-9.061)	(-8.975)	(-8.836)	(-8.985)	
Book-to-Market	-0.029	-0.015	-0.007	-0.028	
	(-0.221)	(-0.120)	(-0.060)	(-0.212)	
ROA	0.249	0.274	0.289	0.253	
	(0.546)	(0.599)	(0.643)	(0.555)	
Leverage	0.614	0.618	0.648	0.629	
	(0.972)	(0.973)	(1.013)	(1.008)	
Relative Deal Size	-0.625***	-0.629***	-0.629***	-0.625***	
	(-6.730)	(-6.755)	(-6.787)	(-6.739)	
Cross-industry	-0.189	-0.188	-0.186	-0.189	
oroso madory	(-1.125)	(-1.115)	(-1.104)	(-1.123)	
Cross-border	-0.747*	-0.733*	-0.733*	-0.748*	
	(-1.817)	(-1.772)	(-1.755)	(-1.797)	
Cash deal	0.344	0.332	0.324	0.343	
out the transfer of the transf	(1.013)	(0.989)	(0.958)	(1.002)	
Private Target	0.098	0.099	0.101	0.099	
Tittate Tanget	(0.168)	(0.169)	(0.173)	(0.168)	
Tender Offer	0.556	0.543	0.551	0.564	
Tender offer	(1.367)	(1.333)	(1.349)	(1.403)	
Acquirer country FE	Yes	Yes	Yes	Yes	
Target country FE	Yes	Yes	Yes	Yes	
Industry FE	Yes	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	Yes	
N N	5322	5322	5322	5322	
Adjusted R ²	0.064	0.063	0.063	0.064	

This table presents the effect of different board reform components on corporate acquisition performance. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. Components of board reforms are indicator variables that equal one for the five years after the related board reforms, and zero for the five years before the related board reforms. Appendix 1 presents detailed definitions and constructions of the variables. t-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

influence over crucial decisions (Finkelstein and D'Aveni, 1994; Adams et al., 2005). Hence, the effect of board reforms involving the separation of the CEO and board chair positions remains an empirical question.

Columns 1 to 3 of Table 6 presents the results that examines how the three types of board reforms individually affect corporate acquisition performance. We find that specific reforms on board independence (Board Independence Reform), but not the requirement of audit committee independence (Audit Committee Reform) nor the separation of CEO and chairman positions (Separation of CEO-Chair Reform), is the main driver of the documented positive effect of board reforms on corporate acquisition performance. Specifically, Column 1 of Table 6 shows that cumulative acquirer abnormal returns on average decreases by 1.3% after the adoption of board independence reform. This finding complements the recent finding by Masulis and Zhang (2019) that firms with more preoccupied independent directors perform worse in acquisitions.

In column 4 of Table 6 we also include all three types of board reforms in one regression to jointly test the effects of these board reforms. First, we find that the coefficient estimate for *Audit Committee Reform* is negative, however it is insignificant and the coefficient estimate for *Separation of CEO-Chair Reform* remains positive. Second and more importantly, the coefficient estimate on *Board Independence Reform* stays positive and significant, suggesting that board independence on average adds value to acquisition decisions.

Table 7Global board reforms and corporate acquisition performance conditional on firm characteristics.

Dependent Variable	CAR $(-2, +2)$	
	(1)	(2)
Reform x Firm Size	0.312***	
	(2.882)	= <00.111
Reform x Free Cash Flow		5.633***
Free Cash Flow		(5.462) -1.801
Free Cash Flow		(-1.000)
Reform	0.088	2.066*
Reform	(0.088)	(1.765)
Ln(GDP)	4.099**	2.179
EII(GDI)	(2.342)	(1.090)
GDP Growth	0.061	0.166
dbi diowai	(0.303)	(0.841)
Target Ln(GDP)	0.946	3.035
Tanget En(GDT)	(0.605)	(1.449)
Target GDP Growth	-0.145	-0.295
Tanger ob 1 drown	(-0.645)	(-1.351)
Firm Size	-0.737***	-0.686***
Time diffe	(-14.432)	(-9.569)
Book-to-Market	-0.046	-0.219*
2001 to Million	(-0.313)	(-1.849)
ROA	0.247	0.114
	(0.564)	(0.073)
Leverage	0.618	0.489
	(0.977)	(1.241)
Relative Deal Size	-0.626***	-0.764***
	(-6.243)	(-9.944)
Cross-industry	-0.168	-0.065
,	(-0.944)	(-0.311)
Cross-border	-0.754*	-0.785*
	(-1.876)	(-2.044)
Cash deal	0.319	0.500
	(1.076)	(1.244)
Private Target	0.111	-0.343
· · · · · · · · · · · · · · · · · · ·	(0.185)	(-0.740)
Tender Offer	0.566	0.478
	(1.363)	(1.442)
Acquirer country FE	Yes	Yes
Target country FE	Yes	Yes
Industry FE	Yes	Yes
Year FE	Yes	Yes
N	5322	3922
Adjusted R ²	0.065	0.067

This table presents the effect of global board reforms on corporate acquisition performance conditional on firm characteristics. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

4. Further analysis

Our results so far show that corporate acquisition performance improves after the implementation of board reforms, which is consistent with agency theory predictions. In this section, we conduct cross-sectional tests to examine in detail how strengthening board independence can alleviate agency problems between managers and shareholders and improve acquisition performance.

4.1. Agency conflicts

We first investigate the types of acquirers that are driving the positive effect of board reforms on acquisition performance ('Board reform strengthening' effect). It is well known in the literature that the market for corporate control helps to discipline poorly performing managers. Mitchell and Lehn (1990), for example, document that firms that perform poorly in acquisitions are at a later date likely to be acquired themselves. However, prior studies show that the market for corporate control is much less effective for large firms, as it is very difficult to acquire a large target (Palepu, 1986; Powell, 1997). Consistent with the evidence that managers in large firms face less threat from the market for corporate control, Moeller et al. (2004) show that large acquirers perform much worse in acquisitions, with announcements returns around 2% lower for large acquirers compared to small acquirers. We therefore expect that

 Table 8

 Global board reforms and corporate acquisition performance conditional on deal characteristics.

Dependent Variable	CAR (-2, +2)	
	(1)	(2)
Reform x Relative Deal Size	-1.230**	
	(-2.232)	
Reform x Private Target		1.222***
C		(18.625)
Reform	2.519**	1.627*
	(2.486)	(1.863)
Ln(GDP)	3.570**	3.609**
	(2.066)	(2.185)
GDP Growth	-0.003	0.034
	(-0.014)	(0.148)
Target Ln(GDP)	1.211	1.115
	(0.772)	(0.758)
Target GDP Growth	-0.132	-0.131
· ·	(-0.575)	(-0.556)
Firm Size	-0.574***	-0.573***
	(-9.432)	(-9.092)
Book-to-Market	-0.032	-0.057
	(-0.242)	(-0.462)
ROA	0.331	0.299
	(0.755)	(0.687)
Leverage	0.637	0.619
	(0.971)	(0.945)
Relative Deal Size	-0.623***	-1.269***
	(-6.591)	(-18.902)
Cross-industry	-0.207	-0.181
,	(-1.221)	(-1.047)
Cross-border	-0.695	-0.718*
	(-1.653)	(-1.721)
Cash deal	0.357	0.341
	(1.096)	(1.074)
Private Target	0.872	0.124
Tittate Target	(1.335)	(0.209)
Tender Offer	0.658	0.579
	(1.470)	(1.385)
Acquirer country FE	Yes	Yes
Target country FE	Yes	Yes
Industry FE	Yes	Yes
Year FE	Yes	Yes
N N	5322	5322
Adjusted R ²	0.065	0.066

This table presents the effect of global board reforms on corporate acquisition performance conditional on deal characteristics. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

managers in larger firms are more likely to conduct self-serving acquisitions when they are not as carefully monitored, and hence the implementation of board reforms should have a stronger effect for larger acquirers. Column 1 of Table 7 examines the effect of board reforms on acquisition performance conditional on acquirer size. We find indeed that the positive effect of board reforms is stronger for larger acquirers.

The level of free cash flow is another strong determinant of acquisition performance documented in the literature. For example, Jensen (1986) argues that managers in firms with more free cash flows are more likely to make value-destroying acquisitions in order to realize personal gains from empire building, rather than returning the excess cash flow to shareholders. Lang et al. (1991) later test and find strong supportive empirical evidence for the free cash flow hypothesis. Consistent with these findings, we expect that managers in firms with higher levels of free cash flows are more likely to conduct self-serving acquisitions if not carefully monitored, and hence the implementation of board reforms should also have a stronger effect for firms with more free cash flows. Column 2 of Table 7 reports the results. Consistent with our prediction, we find that the positive effect of board reforms is stronger for acquirers with more free cash flows.

Table 9
Global board reforms and corporate acquisition performance conditional on investor protection.

Dependent Variable	CAR $(-2, +2)$	
	(1)	(2)
Reform x Poor Shareholder Rights	2.812**	
	(2.254)	
Reform x Poor Creditor Rights		1.220**
		(2.318)
Reform	0.594	0.494
	(0.639)	(0.529)
Ln(GDP)	3.459*	4.003**
	(1.967)	(2.056)
GDP Growth	-0.060	0.049
	(-0.282)	(0.246)
Target Ln(GDP)	1.064	0.600
	(0.736)	(0.399)
Target GDP Growth	-0.104	-0.084
	(-0.486)	(-0.389)
Firm Size	-0.565***	-0.577***
	(-8.951)	(-8.787)
Book-to-Market	0.017	-0.035
	(0.151)	(-0.263)
ROA	0.223	0.231
_	(0.509)	(0.517)
Leverage	0.589	0.645
	(0.896)	(0.985)
Relative Deal Size	-0.633***	-0.636***
	(-7.008)	(-7.423)
Cross-industry	-0.180	-0.184
	(-1.064)	(-1.075)
Cross-border	-0.729*	-0.706*
	(-1.808)	(-1.756)
Cash deal	0.381	0.377
	(1.161)	(1.123)
Private Target	0.133	0.108
	(0.224)	(0.179)
Tender Offer	0.604	0.587
	(1.535)	(1.472)
Acquirer country FE	Yes	Yes
Target country FE	Yes	Yes
Industry FE	Yes	Yes
Year FE	Yes	Yes
N	5167	5167
Adjusted R ²	0.065	0.065

This table presents the effect of global board reforms on corporate acquisition performance conditional on investor protection. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Poor shareholder rights equals to one if the Shareholder rights index for the acquirer country is 0, 1, or 2 (within a total range of 0–5), and zero otherwise. Poor creditor rights equals to one if the Creditor rights index for the acquirer country is 0 or 1 (within a total range of 0–4), and zero otherwise. Appendix 1 presents detailed definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

Next, we investigate the types of acquisition deals that are driving the positive effect of board reforms on acquisition performance. We expect the increase in acquisition performance to be concentrated in deals that are more likely subject to agency conflicts, as the forced addition of independent directors post-reform would play a significant role in monitoring managers in conducting these deals.

It is well documented in the literature that managers have strong incentives to engage in empire-building if not carefully monitored. Studies document that entrenched managers have incentives to grow their firms excessively in order to increase the resources that they control, as growth in resources increases a manager's power and typically also their compensation level (Murphy, 1985; Jensen, 1986; Harford, 1999). In support of this view, Moeller et al. (2004, 2005) document that large acquisitions are more value-reducing. Hence, we expect the positive effect of board reforms on acquisition performance to be driven by larger deals, as the added independent directors post reform are likely to closely monitor larger deals to ensure that these deals are not conducted due to empire-building motives. Column 1 of Table 8 examines the effect of board reforms on acquisition performance conditional upon relative deal size. Consistent with our expectations, we find that the positive effect of board reforms is stronger in deals with larger relative deal size.

The preference for acquiring public targets is another channel documented in the literature through which entrenched managers destroy shareholder value in corporate acquisitions. For example, Harford et al. (2012) show that entrenched managers are likely to avoid acquiring private firms even if it is value enhancing because acquiring a private target can create a large blockholder with the power and incentive to monitor and discipline them going forward. Therefore, we expect the positive effect of board reforms on acquisition performance to also be concentrated in public-target deals, as the public-target acquisitions conducted before the implementation of board reforms may be pursued not with the purpose of maximising shareholder value. Column 2 of Table 8 examines the effect of board reforms on acquisition performance conditional upon target public status. Consistent with our expectations, we find that the positive effect of board reforms is stronger in public-target deals.

We also examine the effect of board reforms on acquisition performance conditional upon cross-industry deals and cash deals. The univariate comparisons in Table 2 show that the proportion of cross industry deals and cash deals increased significantly in the post-reform period, and therefore we investigate whether the positive effect of board reforms on acquisition performance is also driven by these specific types of deals. The results are shown in Appendix 7. We do not find that the positive effect of board reforms on acquisition performance is driven by these types of deals, further suggesting that it is the alleviation of agency conflicts that is driving our results.

4.2. Country-level institutional environment

Next, we examine the role country-level institutional environment plays in affecting the impact of board reforms on corporate acquisition performance. Pinkowitz et al. (2006) find that managers in countries with poor investor protection are more likely to extract private benefits, and Doidge et al. (2007) demonstrate that country characteristics explain significantly more of the variance in corporate governance (39% to 73%) compared to observable firm characteristics (4% to 22%). Due to these prior findings, we expect the positive effect of board reforms on corporate acquisition performance to be concentrated in firms operating in countries with exante poor corporate governance, as the board reforms are likely to play a very important role in disciplining managers in these countries with weaker corporate governance standards. On the other hand, corporate managers in countries with strong corporate governance standards are likely to be restricted to an extent in conducting self-serving acquisitions even prior to the adoption of board reforms, and hence the implementation of board reforms should not have a significant effect on corporate acquisition performance in these countries. To test our predictions, we condition our DID estimations on the level of ex-ante shareholder protection and creditor protection in a country, given that both shareholders and creditors have incentives to prevent managers from conducting self-serving acquisitions.

We use the anti-director rights index introduced by La Porta et al. (1998) as a proxy for shareholder protection. The index ranges from 0 to 5 and it measures how the legal system favors shareholders against managers in the corporate decision-making process, including the voting process. Higher values indicate better shareholder protection. We expect that managers in firms operating in legal systems favoring managers over shareholders are more likely to conduct self-serving acquisitions, since shareholders in these firms are unlikely to have the ability to replace a manager through the voting process. We classify a firm as having a poor level of shareholder protection if they operate in a country with a shareholder protection score of 0, 1, or 2. Column 1 of Table 9 shows the results. We find that the positive effect of board reforms on corporate acquisition performance ('Board reform strengthening') is stronger in firms operating in countries with ex-ante weaker shareholder protection.

Next, we use the creditor rights index used by La Porta et al. (1998) as a measure of creditor protection. The index ranges from 0 to 4 and it measures the power a country's law and regulation provide to secured lenders. We expect that managers of firms that operate in countries providing less power to secured lenders are more likely to conduct self-serving acquisitions, since managers in these firms can continue to run the business even if the firm runs into reorganization, rather than having the creditor or administrator taking over in running the business. We classify a firm as having a poor level of creditor protection if they operate in a country with a creditor protection score of 0 or 1. Column 2 of Table 9 presents the results. Similar to our results on shareholder protection, we also find that the positive effect of board reforms on corporate acquisition performance ('Board reform strengthening') is concentrated in firms operating in countries with weaker creditor protection.

5. Conclusion

This study examines the effect of board reforms on corporate acquisition performance using a geographically diverse sample from around the world. Using a difference-in-differences approach, we show that corporate acquisition performance increases after the implementation of board reforms in the acquirer's country, consistent with a 'Board reform strengthening' view. We find that the

increase is driven by reforms involving board independence, but not reforms involving audit committee independence nor the separation of the CEO and board chair roles (removal of CEO duality). Further analysis reveals that the improvement in acquisition performance ('Board reform strengthening') is concentrated where agency conflicts are most likely - in larger acquirers with more free cash flows, large deals, public-target deals, and in acquirer countries with ex-ante poor shareholder and creditor protection.

Overall, our findings show that the implementation of board reforms alleviates agency problems between managers and share-holders and improves corporate acquisition performance – a finding that should be of interest to investors, academics, and regulators. Future research in this area should focus on whether board reforms can alleviate other types of managerial misbehavior, such as earnings management, corporate fraud, insider trading and tunnelling.

CRediT authorship contribution statement

To Thomas: Conceptualization, Formal analysis, Methodology, Supervision, Validation, Writing – review & editing. **Eliza Wu:** Conceptualization, Methodology, Project administration, Supervision, Visualization, Writing – review & editing. **Diya Zhao:** Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Visualization, Writing – original draft.

Data availability

Data will be made available on request.

Appendix 1: Variable Definitions

Variable name	Variable definition	Source
CAR (-2, +2)	Acquirer's five-day cumulative abnormal return around deal announcement. Abnormal returns are	Compustat
CAR (-1, +1)	calculated by subtracting the acquirer country's stock market return from the acquirer's stock return. Acquirer's three-day cumulative abnormal return around deal announcement. Abnormal returns are calculated by subtracting the acquirer country's stock market return from the acquirer's stock return.	Compustat
ΔIndustry-Adjusted ROA	ΔIndustry-Adjusted ROA is calculated as the acquirer's industry-adjusted ROA in the year after the acquisition minus acquirer's industry-adjusted ROA in the year before the acquisition. Industry-adjusted ROA is measured as acquirer's ROA minus the average ROA of firms in the same two-digit SIC code. ΔIndustry-Adjusted Sales Growth is calculated as the acquirer's industry-adjusted sales growth in the	Compustat
	year after the acquisition minus acquirer's industry-adjusted sales growth in the year before the	
ΔIndustry-Adjusted Sales Growth	acquisition. Industry-adjusted sales growth is measured as acquirer's sales growth minus the average sales growth of firms in the same two-digit SIC code.	Compustat
Ln (GDP)	Natural logarithm of GDP for the acquirer country, in thousands of \$US dollars.	World Development Indicators
GDP Growth	Acquirer country's GDP growth.	World Development Indicators World Development
Target Reform	An indicator variable that equals one for the five years post first board reform in target country, and zero for the five years before first board reform in target country.	Indicators
Target Ln(GDP)	Natural logarithm of GDP for the target country, in thousands of \$US dollars.	World Development Indicators
Target GDP Growth	Target country's GDP growth.	World Development Indicators Compustat
Firm Size	Natural logarithm of market value of equity of the acquirer, in US dollars.	<u>.</u>
	Acquirer's book value of equity (CEQ) divided by market value of equity	Compustat
Book-to-Market		
ROA Leverage	Acquirer's operating income before depreciation (OIBDP) scaled by the book value of total assets (AT). Acquirer's Book value of debt (DLTT+DLC) scaled by the book value of total assets (AT) Acquirer's operating income before depreciation (OIBDP) minus interest expense (XINT) minus income	Compustat Compustat
Free cash flow	taxes (TXT) minus capital expenditure (CAPX), all scaled by book value of total assets (AT).	Compustat
Relative deal size	Deal value (from SDC) divided by acquirer's market value of equity.	Compustat and SDC Platium
Cross-industry	An indicator variable that equals one if acquirer and target do not share the same 2-digit SIC industry, and zero otherwise.	SDC Platium
Cross-border	An indicator variable that equals one if the target nation is not the same as acquirer nation, and zero otherwise.	SDC Platium
Private Target	An indicator variable that equals one for private targets, and zero otherwise. An indicator variable that equals one for tender offer, and zero otherwise.	SDC Platium
Tender Offer	Shareholder rights index of the acquirer's country.	SDC Platium
Shareholder Rights		La Porta et al. (1998)
		(continued on next pa

Variable name	Variable definition	Source
Creditor Rights	Creditor rights index of the acquirer's country.	La Porta et al. (1998)
Takeover Premium	Difference between the offer price and the target market value 1 week prior to the announcement.	SDC

This table provides the definitions and data sources of all the variables used in the study.

Appendix 2: Types of Board Reforms

Country	Year	Board Independence	Separation of CEO and Chairman	Audit Committee Independence
Australia	2003	1	0	1
Belgium	1998	1	0	0
Brazil	2002	0	0	0
Canada	2004	1	1	1
China	2001	1	0	1
Colombia	2001	0	0	0
Denmark	2001	1	0	0
Finland	2003	1	0	1
France	2001	0	1	0
Germany	2002	1	0	1
Greece	1999	0	0	0
Hong Kong (China)	2005	1	1	1
India	1998	0	0	0
Indonesia	2000	1	0	0
Italy	2006	1	0	1
Japan	2002	0	0	1
Malaysia	2001	1	0	1
Mexico	1999	1	0	1
Netherlands	1997	0	0	0
Norway	2005	1	1	1
Peru	2002	1	1	1
Philippines	2002	1	0	1
Poland	2002	1	0	0
Portugal	1999	0	0	0
Singapore	2003	1	0	1
South Korea	1999	1	0	1
Spain	1998	0	0	1
Sweden	2005	1	1	1
Switzerland	2002	0	0	0
Thailand	2002	1	0	1
United States	2003	1	0	1

This table contains information on sample countries' first board reform. It includes the year of the reform and the component the reform involves (i.e., board independence, separation of CEO and chairman, and audit committee independence).

Appendix 3: Global Board Reforms and Corporate Acquisition Performance: Alternative Sample

Dependent variable	CAR (-2, +2)			
	(1)	(2)	(3)	
	Owns less than 10% of target before the announcement	Owns less than 20% of target before the announcement	Owns less than 50% of target before the announcement	
Reform	1.832**	1.779**	1.421*	
	(2.077)	(2.070)	(1.832)	
Ln(GDP)	3.427**	3.743**	4.415***	
	(2.151)	(2.398)	(3.083)	
GDP Growth	0.013	0.018	-0.026	
	(0.058)	(0.079)	(-0.123)	
Target Ln(GDP)	1.227	1.143	0.101	
	(0.852)	(0.851)	(0.068)	
Target GDP Growth	-0.146	-0.153	-0.108	
	(-0.634)	(-0.651)	(-0.517)	
Firm Size	-0.569***	-0.559***	-0.553***	

(continued on next page)

Dependent variable	CAR (-2, +2)			
	(1)	(2)	(3)	
	(-9.185)	(-9.408)	(-8.424)	
Book-to-Market	-0.024	-0.022	-0.036	
	(-0.184)	(-0.172)	(-0.304)	
ROA	0.309	0.220	0.246	
	(0.704)	(0.530)	(0.635)	
Leverage	0.641	0.637	0.559	
	(1.105)	(1.215)	(1.102)	
Relative Deal Size	-0.628***	-0.628***	-0.629***	
	(-6.342)	(-6.607)	(-5.923)	
Cross-industry	-0.220	-0.242	-0.221	
	(-1.302)	(-1.433)	(-1.455)	
Cross-border	-0.737*	-0.637	-0.442	
	(-1.758)	(-1.567)	(-1.232)	
Cash deal	0.344	0.355	0.237	
	(1.057)	(1.055)	(0.710)	
Private Target	0.104	0.131	0.101	
	(0.179)	(0.238)	(0.194)	
Tender Offer	0.506	0.528	0.529	
	(1.098)	(0.979)	(1.038)	
Acquirer country FE	Yes	Yes	Yes	
Target country FE	Yes	Yes	Yes	
Industry FE	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	
N	5351	5455	5893	
Adjusted R ²	0.063	0.061	0.058	

This table presents the effect of global board reforms on corporate acquisition performance using alternative samples. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 provides the definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

Appendix 4: Global Board Reforms and Corporate Acquisition Performance: Alternative CAR (-2, +2)

Dependent Variable	CAR $(-2, +2)$			
	Full Sample		Exclude countries with less than 10 unique acquirers	
Reform	0.023**	0.025**	0.026**	
	(2.126)	(2.327)	(2.311)	
Ln(GDP)	0.010	0.018	0.017	
	(0.547)	(0.874)	(0.782)	
GDP Growth	0.001	0.000	0.000	
	(0.296)	(0.239)	(0.198)	
Target Ln(GDP)	0.016	0.014	0.016	
_	(0.781)	(0.667)	(0.733)	
Target GDP Growth	-0.002	-0.002	-0.002	
0	(-1.383)	(-1.317)	(-1.297)	
Firm Size	-0.006***	-0.005***	-0.005***	
	(-11.314)	(-5.404)	(-5.156)	
Book-to-Market	0.002	0.002	0.002	
	(1.186)	(1.341)	(1.351)	
ROA	0.021***	0.018**	0.020**	
	(2.925)	(2.177)	(2.169)	
Leverage	-0.001	0.001	0.001	
o .	(-0.256)	(0.108)	(0.132)	
Relative Deal Size		-0.007***	-0.007***	
		(-6.129)	(-7.342)	
Cross-industry		-0.001	-0.001	
•		(-0.837)	(-0.800)	
Cross-border		-0.003	-0.003	
		(-0.612)	(-0.597)	
Cash deal		0.006	0.005	
		(1.255)	(1.106)	

(continued on next page)

Dependent Variable	CAR(-2, +2)			
	Full Sample		Exclude countries with less than 10 unique acquirers	
Private Target		-0.001	-0.000	
		(-0.098)	(-0.018)	
Tender Offer		0.010*	0.010*	
		(1.866)	(1.977)	
Acquirer country FE	Yes	Yes	Yes	
Target country FE	Yes	Yes	Yes	
Industry FE	No	Yes	Yes	
Year FE	Yes	Yes	Yes	
N	4933	4933	4893	
Adjusted R ²	0.055	0.060	0.057	

This table presents the effect of global board reforms on corporate acquisition performance with alternative standard error clustering. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. The CARs are measured as returns in excess of those predicted by the market model with the benchmark being the acquiring country's index and parameters estimated over a period from 300 to 91 days prior to the announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and constructions of the variables. t-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

Appendix 5: Global Board Reforms and Corporate Acquisition Performance: Alternative Standard Error Clustering

Dependent Variable	CAR $(-2, +2)$			
	(1)	(2)	(3)	
	Cluster by Acquirer	Cluster by Industry	Cluster by Year	
Reform	1.941*	1.941**	1.941***	
	(1.895)	(2.142)	(3.180)	
Ln(GDP)	3.434*	3.434	3.434*	
	(1.716)	(1.047)	(1.931)	
GDP Growth	0.001	0.001	0.001	
	(0.007)	(0.007)	(0.008)	
Target Ln(GDP)	1.319	1.319	1.319	
	(0.620)	(0.726)	(0.656)	
Target GDP Growth	-0.138	-0.138	-0.138	
g	(-0.855)	(-0.905)	(-0.992)	
Firm Size	-0.569***	-0.569***	-0.569***	
	(-7.002)	(-5.705)	(-8.294)	
Book-to-Market	-0.023	-0.023	-0.023	
	(-0.186)	(-0.191)	(-0.163)	
ROA	0.256	0.256	0.256	
	(0.226)	(0.261)	(0.281)	
Leverage	0.599	0.599	0.599	
	(0.779)	(0.867)	(0.847)	
Relative Deal Size	-0.631***	-0.631***	-0.631***	
readive Bear Size	(-3.426)	(-4.005)	(-3.011)	
Cross-industry	-0.194	-0.194	-0.194	
Gross madely	(-0.677)	(-0.765)	(-0.734)	
Cross-border	-0.743	-0.743	-0.743	
Gross Border	(-1.306)	(-1.596)	(-1.383)	
Cash deal	0.350	0.350	0.350	
Cusir dear	(1.280)	(1.012)	(1.398)	
Private Target	0.096	0.096	0.096	
Tilvate Target	(0.348)	(0.411)	(0.306)	
Tender Offer	0.551	0.551	0.551	
render oner	(1.071)	(1.130)	(0.767)	
Acquirer country FE	Yes	Yes	Yes	
Target country FE	Yes	Yes	Yes	
Industry FE	Yes	Yes	Yes	
Year FE	Yes	Yes	Yes	
N	5322	5322	5322	
Adjusted R ²	0.064	0.064	0.064	

This table presents the effect of global board reforms on corporate acquisition performance with alternative standard error clustering. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and

constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

Appendix 6: Global Board Reforms and Takeover Premium

Dependent Variable	Takeover Premium
	(1)
Reform	10.312
	(1.194)
Ln(GDP)	-37.949*
	(-1.905)
GDP Growth	2.161
	(0.825)
Target Ln(GDP)	19.516
	(1.373)
Target GDP Growth	-1.790
_	(-1.187)
Firm Size	-1.348***
	(-3.104)
Book-to-Market	-2.859**
	(-2.839)
ROA	-7.972
	(-1.207)
Leverage	6.044***
-	(5.319)
Relative Deal Size	-1.075***
	(-5.606)
Cross-industry	-4.854**
	(-2.915)
Cross-border	14.916***
	(3.542)
Cash deal	5.301***
	(3.966)
Private Target	-12.778
	(-1.253)
Tender Offer	4.746***
	(4.156)
Acquirer country FE	Yes
Target country FE	Yes
Industry FE	Yes
Year FE	Yes
N	1321
Adjusted R ²	0.162

This table presents the effect of global board reforms on takeover premium. The dependent variable, takeover premium, is the percentage difference between the offer price and the target's market value 1 week prior to the announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 provides the definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

Appendix 7: Global Board Reforms and Corporate Acquisition Performance conditional on Cross-Industry Deals and Cash Deals

Dependent Variable	CAR (-2,+2)		
	(1)	(2)	
Reform x Cross-Industry	-0.084 (-0.385)		
Reform x Cash deal		-0.170	
		(-0.344)	
	(continued on next page)	

Adams, R.B., Ferreira, D., 2007. A theory of friendly boards. J. Financ. 62, 217-250.

Duchin, R., Matsusaka, J., Ozbas, O., 2010. When are outside directors effective? J. Financ. Econ. 96, 195–214. Faleye, O., Hoitash, R., Hoitash, U., 2011. The costs of intense board monitoring. J. Financ. Econ. 101, 160–181.

Fama, E.F., Jensen, M.C., 1983. Separation of ownership and control. J. Law Econ. 26, 301-325.

pendent Variable	CAR $(-2,+2)$		
	(1)	(2)	
form	1.971**	1.971**	
	(2.184)	(2.149)	
(GDP)	3.429*	3.402*	
	(2.028)	(2.037)	
OP Growth	0.003	0.002	
	(0.015)	(0.009)	
rget Ln(GDP)	1.313	1.322	
	(0.860)	(0.867)	
rget GDP Growth	-0.137	-0.137	
	(-0.592)	(-0.584)	
rm Size	-0.570***	-0.568***	
	(-8.968)	(-8.645)	
ok-to-Market	-0.024	-0.022	
	(-0.183)	(-0.173)	
)A	0.254	0.245	
	(0.573)	(0.533)	
verage	0.600	0.590	
	(0.934)	(0.928)	
lative Deal Size	-0.631***	-0.629***	
	(-6.853)	(-7.109)	
oss-industry	-0.142	-0.194	
	(-0.685)	(-1.140)	
oss-border	-0.745*	-0.741*	
	(-1.820)	(-1.785)	
sh deal	0.350	0.466	
	(1.069)	(0.900)	
ivate Target	0.094	0.096	
Tute Turget	(0.161)	(0.164)	
nder Offer	0.550	0.531	
nder oner	(1.351)	(1.408)	
quirer country FE	Yes	Yes	
rget country FE	Yes	Yes	
dustry FE	Yes	Yes	
ar FE	Yes	Yes	
	5322	5322	
linsted R ²		0.064	
justed R ²	0.064		

This table presents the effect of global board reforms on corporate acquisition performance conditional on cross-industry deals and cash deals. The dependent variable CAR (-2, +2) is the five-day cumulative abnormal return around the acquisition announcement. *Reform* is an indicator variable equal to one for the five years after the first board reform of the acquirer country, and zero for the five years before the first board reform of the acquirer country. Appendix 1 presents detailed definitions and constructions of the variables. *t*-statistics, reported in parentheses, are based on standard errors clustered at the country level. ***, **, * denote significance levels at 1%, 5% and 10% respectively.

References

```
Adams, R.B., Almeida, H., Ferreira, D., 2005. Powerful CEOs and their impact on corporate performance. Rev. Financ. Stud. 18, 1403–1432.

Bae, K., Ghoul, S.E., Guedhami, O., Zheng, X., 2021. Board reforms and dividend policy: international evidence. J. Financ. Quant. Anal. 56, 1296–1320.

Baker, A., Larcker, D.F., Wang, C.C., 2021. How much should we trust staggered differences estimates?. In: Harvard Business School Working Paper, No. 21-112.

Ben-David, I., Bhattacharya, U., Jacobsen, S., 2022. Announcement returns and value creation. In: NBER Working Paper No. 27976.

Ben-Nasr, H., Boubaker, S., Sassi, S., 2021. Board reforms and debt choice. Finance 69, 102009.

Bertrand, M., Mullainathan, S., 2003. Enjoying the quiet life? Corporate governance and managerial preferences. J. Polit. Econ. 111, 1043–1075.

Byrd, J.W., Hickman, K.A., 1992. Do outside directors monitor managers? Evidence from tender offer bids. J. Financ. Econ. 32, 195–221.

Carcello, J.V., Neal, T.L., 2002. Audit committee composition and auditor reporting. Account. Rev. 75, 453–467.

Chen, R., Guedhami, O., Yang, Y., Zaynutdinova, G., 2020. Corporate governance and cash holdings: evidence from worldwide board reforms. Finance 65, 101771.

Chen, Y., Goyal, A., Zolotoy, L., 2022. Global board reforms and the pricing of IPOs. J. Financ. Quant. Anal. 57, 2412–2443.

Deli, D., Gillian, S., 2000. On the demand for independent and active audit committees. Finance 6, 427–445.

Djankov, S., La Porta, R., Lopez-de-Silanes, F., Shleifer, A., 2008. The law and economics of self-dealing. J. Financ. Econ. 88, 430–465.

Doidge, C., Karolyi, A., Stulz, R., 2007. Why do countries matter so much for corporate governance? J. Financ. Econ. 86, 1–39.
```

Fauver, L., Hung, M., Li, X., Taboada, A.G., 2017. Board reforms and firm value: worldwide evidence. J. Financ. Econ. 125, 120-142.

Finkelstein, S., D'Aveni, R.A., 1994. CEO duality as a double-edged sword: how boards of directors balance entrenchment avoidance and unity of command. Acad. Manag, J. 37, 1079-1108.

Golubov, A., Yawson, A., Zhang, H., 2015. Extraordinary acquirers. J. Financ. Econ. 116, 314-330.

Gormley, T.A., Matsa, D.A., 2016. Playing it safe? Managerial preferences, risk, and agency conflicts. J. Financ. Econ. 122, 431-455.

Guo, L., Masulis, R.W., 2015. Board structure and monitoring: new evidence from CEO turnovers. Rev. Financ. Stud. 28, 2770-2811.

Harford, J., 1999. Corporate cash reserves and acquisitions. J. Financ. 54, 1969–1997.

Harford, J., Humphery-Jenner, M., Powell, R., 2012. The sources of value destruction in acquisitions by entrenched managers. J. Financ. Econ. 106, 247-261. Harris, M., Raviy, A., 2008. A theory of board control and size. Rev. Financ. Stud. 21, 1797-1832.

Hermalin, B., Weisbash, M., 2003. Boards of directors as an endogenously determined institution: A survey of the economic literature. Econ. Policy Rev. 9, 7-26.

Hu, J., Li, S., Taboada, A.G., Zhang, F., 2020. Corporate board reforms around the world and stock price crash risk. Finance 62, 101557.

Hui, C., Lin, C., Wei, L., 2023. How do Board reforms affect debt financing costs around the world? J. Financ. Quant. Anal, 58, 217-249.

Jensen, M.C., 1986. Agency costs of free cash flow, corporate finance, and takeovers. Am. Econ. Rev. 76, 323-329.

Jensen, M.C., 1993. The modern industrial revolution, exit, and the failure of internal control systems. J. Financ. 48, 831-880.

Kim, E.H., Lu, Y., 2013. Corporate governance reforms around the world and cross-border acquisitions. Finance 22, 236-253.

Klein, A., 2002. Audit committee, board of director characteristics, and earnings management. J. Account. Econ. 33, 375-400.

La Porta, R., Lopez-de-Silanes, F., Shleifer, A., Vishny, R.W., 1998. Law and finance. J. Polit. Econ. 106, 1113-1155.

Lang, L., Stulz, R., Walkling, R.A., 1991. A test of the free cash flow hypothesis: the case of bidder returns. J. Financ, Econ. 29, 315-335. Leuz, C., Nanda, D., Wysocky, P., 2003. Earnings management and investor protection: an international comparison. J. Financ. Econ. 69, 505-527.

Li, M., Wu, J., Zhang, L., Zou, L., 2020. Board reforms and the cost of equity: international evidence. Account. Finance 60, 4497-4531.

Masulis, R.W., Mobbs, S., 2011. Are all inside directors the same? Evidence from the external directorship market. J. Financ. 66, 823-872.

Masulis, R.W., Zhang, E.J., 2019. How valuable are independent directors? Evidence from external distractions. J. Financ. Econ. 132, 226-256.

Masulis, R.W., Wang, C., Xie, F., 2007. Corporate governance and acquirer returns. J. Financ. 62, 1851-1889.

Mitchell, M.L., Lehn, K., 1990. Do bad bidders become good targets? J. Polit. Econ. 98, 372–398.

Mitchell, M., Mulherin, J.H., 1996. The impact of industry shocks on takeover and restructuring activity. J. Financ. Econ. 41, 193-229.

Moeller, S.B., Schlingemann, F.P., 2005. Global diversification and bidder gains: a comparison between cross-border and domestic acquisitions. J. Bank. Financ. 29, 533-564

Moeller, S.B., Schlingemann, F.P., Stulz, R., 2004. Firm size and the gains from acquisitions, J. Financ. Econ. 73, 201-228.

Moeller, S.B., Schlingemann, F.P., Stulz, R., 2005. Wealth destruction on a massive scale? A study of acquiring-firm returns in the recent merger wave. J. Financ. 60,

Murphy, K.J., 1985. Corporate performance and managerial remuneration: an empirical analysis. J. Account. Econ. 7, 11-42.

Nguyen, B.D., Nielsen, K.M., 2010. The value of independent directors; evidence from sudden deaths, J. Financ, Econ. 98, 550-567.

Palepu, K.G., 1986. Predicting takeover targets: a methodological and empirical analysis. J. Account. Econ. 8, 3-35.

Pinkowitz, L., Stulz, R., Williamson, R., 2006. Does the contribution of corporate cash holdings and dividends to firm value depend on governance? A cross-country analysis. J. Financ. 61, 2725-2751.

Powell, R., 1997. Modelling takeover likelihood. J. Bus. Financ. Acc. 24, 1009–1030.

Raheja, C.G., 2005. Determinants of board size and composition: a theory of corporate boards. J. Financ. Quant. Anal. 40, 283-306.

Rhodes-Kropf, M., Viswanathan, S., 2004. Market valuation and merger waves. J. Financ. 59, 2685-2718.

Rossi, S., Volpin, P., 2004. Cross-country determinants of mergers and acquisitions. J. Financ. Econ. 74, 277-304.

Suk, I., Wang, M., 2021. Does target firm insider trading signal the target's synergy potential in mergers and acquisitions? J. Financ. Econ. 142, 1155-1185.